Author’s response to reviews

Title: The influence of climatic factors on the development of hemorrhagic fever with renal syndrome and leptospirosis during the peak season in Korea: an ecologic study

Authors:

Yadav Joshi (yadavjoshi@gmail.com)
Eun-Hye Kim (dodendlqtnf@gmail.com)
Hae-Kwan Cheong (hkcheong@skku.edu)

Version: 1  Date: 19 Aug 2016

Author’s response to reviews:

Dear Editor,

I have revised the dataset to anonymize individual by transforming the age into a categorical variable of age group.

Thank you.